Title: Is plasma vitamin C an appropriate biomarker of vitamin C intake? A systematic review and meta-analysis

Authors:

Mahshid Dehghan (mahshid@ccc.mcmaster.ca)
Noori Akhtar-Danesh (daneshn@mcmaster.ca)
Catherine R. McMillan (Catherine.McMillan@sunnybrook.ca)
Lehana Thabane (ThabanL@mcmaster.ca)

Version: 4 Date: 1 November 2007

Author's response to reviews:

Dear Editor,

Is plasma vitamin C an appropriate biomarker of vitamin C intake? A systematic review and meta-analysis

On behalf of my co-authors and myself, I would like to thank you once again for the publication of this article in the Nutrition Journal.

We have accommodated all the requested formatting changes in the article and now it is in its final format.

Sincerely,

Mahshid Dehghan